

Thank you for your comments.

All the changes requested have been addressed.

1. Please notice that the name of the journal is Orthopaedic not gastroenterology.
2. All authors' contributions have been added.
3. The results and the conclusion in the abstract have been changed as instructed.
4. The reference citation has been changed.
5. PMID added to each reference.

DEPARTMENT OF ORTHOPAEDIC SURGERY

Dowling Hall, Room 2457

3065 Arlington Avenue

Toledo, Ohio 43614-5807

Phone: 419.383.3515

Fax: 419.383.3526

Hosseini, Elgafy, M.D.MCH, FRCSED, FRCSC

Assistant Professor

Spine Surgery